Author's response to reviews

Title: Electrical short-circuit in beta-cells from a patient with non-insulinoma pancreatogenous hypoglycemic syndrome (NIPHS): A case report

Authors:

Robert Bränström (robert.branstrom@ki.se)
Erik Berglund (Erik.Berglund@ki.se)
Pontus Curman (pontus.curman@sodersjukhuset.se)
Lars Forsberg (larsforsberg65@gmail.com)
Anders Höög (anders.hoog@karolinska.se)
Lars Grimelius (lars.grimelius@genpat.uu.se)
Per-Olof Berggren (Per-Olof.Berggren@ki.se)
Per Mattsson (Per.Mattsson@karolinska.se)
Per Hellman (per.hellman@surgsci.uu.se)
Lisa Juntti-Berggren (Lisa.Juntti-Berggren@ki.se)

Version: 4 Date: 15 June 2010

Author's response to reviews:

Dear editor,

We have made changes as suggested. More specifically;

"Affiliations: Each affiliation should be written in full, in the following format: Department, Institute, (Street), City, (Postal code), Country. This may mean repeating institutional addresses if the authors are from different departments within an institution."

Response: Corrected on page 1.

"Abstract: Please ensure that the abstract includes the patients ethnicity, age and gender. Please remember to update both the manuscript and the submission system so that both abstracts remain identical."

Response: Corrected and updated in manuscript as well as submission system.

"Authors’ contributions: Please indicate that all authors read and approved the final manuscript."

Response: Corrected on page 7.

"References: Please make sure references are cited in numerical order. We notice reference 8 is not cited within the main text. Please make sure references continue to be cited in numerical order once this reference is cited."

Response: Correct.
"Minor revisions (We can make these changes on resubmission, although it will speed up publication of your manuscript if you do them while making the major changes above).

Email address: please include email addresses for all the authors on the title page, using the following format:

Response: e-mail address inserted on title page.

"Coloured text: Please change all text to black."

Response: corrected.

"Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance."

Response: we have made no addition changes.

Yours sincerely,

Robert Bränström
MD, PhD, Assoc professor